Author’s response to reviews

Title: Elucidating mechanistic insights into drug action for atopic dermatitis: a systems biology approach

Authors:

Indhupriya Subramanian (indhupriya_s@persistent.co.in)
Vivek Singh (vivek_ksingh@persistent.com)
Abhay Jere (abhay_jere@persistent.com)

Version: 1 Date: 19 Jul 2017

Author’s response to reviews:

Please kindly prepare your manuscript according to the submission guideline, http://bmcdermatol.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article. The "Results" and "Discussion" should be arranged in two sections, and a "Conclusion" section should be arranged.

Response: We have revised the manuscript according to the submission guidelines. Our manuscript currently has separate sections for "Results", "Discussion" and "Conclusion".

In accordance with BioMed Central editorial policies and formatting guidelines, all manuscript submissions to BMC Dermatology must contain a Declarations section which includes the mandatory sub-sections listed below.

Response: Declarations along with mandatory sub-sections and appropriate information are provided in the manuscript.